Author’s response to reviews

Title: Spectral domain optical coherence tomography in patients after successful management of postoperative endophthalmitis following cataract surgery by pars plana vitrectomy

Authors:

Otto A Maneschg (ottomaneschg@gmail.com)
Éva Volek (volekeva@gmail.com)
János Németh (nemeth.janos@med.semmelweis-univ.hu)
Gábor M Somfai (somfaigm@yahoo.com)
Zsuzsanna Géhl (zsuzsannagehl@gmail.com)
Irén Szalai (szalaiiren@gmail.com)
Miklós D Resch (miklosresch@gmail.com)

Version: 4
Date: 26 May 2014

Author’s response to reviews:

Dear Editor,

We would like to thank you for your quick positive response on our manuscript MS: 1877741433123193.

In the attached manuscript we corrected the questionable points, the p values are checked as suggested. We ensure that the p values are consistent in the manuscript. For correlation between choroidal thickness and central retinal thickness we confirmed the p value > 0.05 and was not considered significant. Exact p value was for study eyes 0.136, for fellow eyes 0.714.

We are grateful for your support and hope that the mentioned corrections in the manuscript are sufficient. Please let us know, if any further information, correction or explanation are necessary.

Yours sincerely

Otto Maneschg, MD

corresponding author

Department of Ophthalmology

Semmelweis University

Budapest, Hungary